Notes of the RICH1 Phone meeting 31/08/2007
People Present:
CERN:  Fabio, Christophe(part), Werner(part)
IC:         Bill(from home), Dave, Richard
Oxford:  Rohan
Bristol:   Nick
Apologies for absence:
Neville, Jon, Olav
Discussed Agenda Topics:
1. Gas Enclosure
Some concern that the dry air supply is introducing dust, although there is no evidence so far. Dowty seals are at CERN. These need checking if they mate with Tim’s plugs (action Fabio).  The temperature sensors connectors will be attached next week by Richard P.
Preparation for the magnet tests (first half of October):

Before the magnet tests the final mirror adjustment should be completed (~mid Sept.). Then the two doors must be put on (~1 day) and a detailed leak test (~2 days) must be done. 

RICH1 survey requirements during magnetic field tests:
Survey the movements of RICH1 with B-field on/off. There are 8 survey target holes on RICH1 (4 on A-side and 4 on C-side). The theodolite may not operate in the magnetic field, and therefore not all survey target holes can be used in the survey. Plan to use any subset of target holes. A 0.5mm survey precision is good enough. Next Wed. meeting held with surveyors at the pit to discuss the different subdetector needs (Fabio will attend).
Clock gauges will be placed on the upper and lower shields as last time to monitor the mechanical movements of the shield (measured w.r.t. the concrete). 
Gas Lines:

The RICH2 gas lines are now finished. The RICH1 gas lines can now be started. In RICH1 there are 4 top and 4 bottom inlets/outlets. The baseline plan is to use only two input and two output because the RICH1 volume is small. A sketch will be prepared and distributed around for discussion (action Dave).

Light shield:

The Milano team installed the upper and lower carbon fibre light shields in front of the VELO exit window.

2. Mirrors
The survey report indicates that small final corrections to the mirror alignment are necessary as expected:
1) The upper flat mirror plane is 248.7mrad (-1.3mrad): need to increase the adjuster length by ~ 0.7mm.

2) The lower flat mirror plane is 253.2mrad (+3.2mrad): need to decrease the adjuster length by ~1.3mm.

3) The CoC is ~10mm too low, needs lifting by tilting back frame by ~ 4.7mm

4) The spherical mirror frame needs to be raised by around ~1mm by adding shims to the rail mounts. 
Next Friday these adjustments will be made. The survey team will come to measure the alignments the following week on the 10th and 11th Sept. 

3. Aerogel
No news. Following the meeting Clara expressed the wish to do a trial installation with the mirrors in place. This was completed successfully by the Milano team on 6 September. 
4. HPD
Column Dressing:  Finishing off spare columns ~2 weeks.
Upper Rail System:  Cross member redesigned because was interfering with the chariot, and the upstream cable system has been relocated. The cross member must go back to NAB for modification (~few weeks). The cable system modification can be done at Oxford. Next Tuesday Rohan will be at IC and will circulate drawings (action Rohan).
Lower Rail System: No news.
Mechanics:  Working on the hood of the HPDs. The drawings should be ready for the workshop by the end of next week.
Cabling: No news.
Light Leak detector:  Two pairs of PMs on order. These will cover a broad spectral range and are of the photon counting type.
Temperature Monitoring: It will be similar to the G.E system. Also a humidity sensor will be included.

Time Alignment laser: Richard is working on it. 

Photon Funnel:  No news
5. HPD Cooling
A solid flexible manifold scheme adopted. The polymer-lined stainless steel flexible pipes can not be used because the lining is incompatible with the cooling liquid.

6. LAMS

Will Reece (IC) is working on a test setup to get a spot with the laser onto the camera. One of the present limitations is the coupling of the laser to the fiber. The system should be ready after the magnet tests in Oct07. 
7. MDMS

Ray has all the mechanics in hand and is working on the optical system to produce small spots. He is using a thick plate with 1mm holes drilled into it. These are capped to form ~0.3mm holes to absorbed reflected light from the inside of the hole. Phone meeting with Ray planned for next week.
Progress report from Ray

*************************************************************************

My main concentration during the past two weeks has been on

the optical design.  I have tested several different designs,

all variations on the idea of using only collimation.  The purpose

of this focus is to keep the optics simple, and not have to resort

to arrays of sub-millimeter size lenses -- unless we have to!

Our original design of consecutive plates of arrays of small holes

does not pass enough light.  The most promising design is a set

of drilled holes in black Delrin (1 mm diam hole from one side,

and 0.3 mm hole from the other side).  The spot size is still not

quite small enough yet, but we've localized the difficulty to a

large component of the light coming at high angles.  Apparently

the drilled hole bore is more reflective than one would naively

expect it to be.  So, the sides of the hole need to be roughened,

and to do this we will try tapping the hole.  This is under test

as I write this.

********************************************************************************************
8. Quartz Windows
No news.
9. Next meeting
Friday 28th September 2007 at 10:00(GMT) 11:00(CET). 
